Author’s response to reviews

Title: Gaucher disease: Single gene molecular characterization of one-hundred Indian patients reveals novel variants and the most prevalent mutation.

Authors:

Jayesh Sheth (jshethad1@gmail.com)
Riddhi Bhavsar (riddhib4@gmail.com)
Mehul Mistri (mistrimehul@yahoo.co.in)
Dhairya Pancholi (riahp7261@gmail.com)
Ashish Bavdekar (ashish.bavdekar1@gmail.com)
Ashwin Dalal (ashwindalal@gmail.com)
Prajnya Ranganath (pragnya4@gmail.com)
Giirisha Katta (girish.katta@manipal.edu)
Anju Shukla (anju.shukla@manipal.edu)
Shubha Phadke (shubharaophadke@gmail.com)
Ratna Puri (ratnadpuri@yahoo.com)
Inusha Panigrahi (inupan@yahoo.com)
Anupriya Kaur (anukaur-genetics@gmail.com)
Mamta Muranja (muranjanmamta@gmail.com)
Manisha Goyal (manidr2000@gmail.com)
Radha Ramadevi (radharamadevi@gmail.com)
Raju Shah (rajucshah@gmail.com)
Sheela Nampoothiri (sheeladr@gmail.com)
Sumita Danda (sumita_danda@hotmail.com)
Chaitanya Datar (dr.cdatar@gmail.com)
Author’s response to reviews:

Cover letter

8th January 2019
To,
The Editor
BioMed Central, UK

Sub.: Resubmission of the manuscript (ID: MGTC-D-18-00440:)

Dear Matteo Pasini,

Enclosed is the manuscript entitled, ‘Gaucher disease: Single gene molecular characterization of one-hundred Indian patients reveals novel variants and the most prevalent mutation.’ for publication in your esteemed journal.

I thank you and for providing valuable inputs in our manuscript. I have uploaded a clean manuscript on the BMC Medical Genetics portal. Should you have any query feel free to contact us. On behalf of all the authors, I confirm that all the authors have gone through the revised manuscript and none has any objection in getting it published in the journal BMC Medical Genetics. The authors declare that they have no competing interests (financial or non-financial) in the present study.

Thanking you,

Yours sincerely,

Dr. Jayesh J. Sheth
FRIGE’s Institute of Human Genetics, FRIGE House, Satellite Ahmedabad-380 015
Gujarat, India.
Tele: +91-79-26921414
Fax: +91-79-26921415
E-mail: jshethad1@gmail.com